Author's response to reviews

Title: Tomophobia, the phobic fear caused by an invasive medical procedure: an upcoming anxiety disorder: a case report.

Authors:

Markus M Schmid (markusmschmid@gmx.de)
Robert C Wolf (christian.wolf@uni-ulm.de)
Roland W Freudenmann (roland.freudenmann@uni-ulm.de)
Carlos Schoenfeldt-Lecuona (carlos.schoenfeldt@uni-ulm.de)

Version: 9 Date: 22 March 2009

Author's response to reviews: see over
Tomophobia, the phobic fear caused by an invasive medical procedure: an upcoming anxiety disorder? A case report

Dear Dr. Kidd,

Thank you for giving us the second opportunity to submit a revised version of our manuscript. We have considered all comments and questions rose by the reviewers and revised the manuscript accordingly. Please find enclosed our reply considering the reviewer’s queries.

We hope that the revised version of our paper is to your full contentment and that the manuscript now qualifies for publication in Journal of Medical Case Reports.

Reply to reviewer:
Thank you for the creative and helpful advices and comments. We have changed all errors and deficiencies following your notes and recommendations. All grammatical errors and formal errors have been revised.
Introduction: P2 Sentence 3 has been changed following your suggestion, as well as P4 Sentence 2 -5.
Discussion: The sentences 6+7 of P1 have been moved to case presentation. The section in P2 last three sentences have been changed completely.
Conclusion: Sentences 2 and 3 have been cut.

We are looking forward to hearing from you.

Yours sincerely

Dr. Markus Schmid